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This morning visited Sodar/Rass everything looked good. Solar was at approx 82%. Also went to PAM Base , ceilometer and all sky camera were working.

At the ISS site 449 and surface met are working . Power on 449 is .126 mW. Will be turning profiler RASS back on this afternoon.

Light breeze. Mostly clear skies. Some clouds to west..

Replaced Voyager back up drive.

Drove out the demons.

Checked fuel reserve tank , should be enough to run till end of project. Will check it each time generator is refilled. Generators should not need to be 
served again during the project

Clayton is checking on the lidar.

RASS mode restarted at 21:45 gmt.
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